Rapid evaluation of the Special Measures for Quality and challenged provider regimes: a mixed-methods study

Naomi J Fulop,^{1*} Estela Capelas Barbosa,² Melissa Hill,¹ Jean Ledger,¹ Pei Li Ng,¹ Christopher Sherlaw-Johnson,³ Lucina Rolewicz,³ Laura Schlepper,³ Jonathan Spencer,³ Sonila M Tomini,¹ Cecilia Vindrola-Padros¹ and Stephen Morris⁴

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/GQQV3512.

Primary conflicts of interest: Naomi J Fulop is a National Institute for Health and Care Research (NIHR) senior investigator and was a member of the following: NIHR Health and Social Care Delivery Research (HSDR) Programme Funding Committee (2013–18) and HSDR Evidence Synthesis Sub Board (2016). Stephen Morris was formerly a member of the NIHR HSDR Programme Funding Committee (2014–19), the NIHR HSDR Evidence Synthesis Sub Board (2018), the NIHR Unmet Need Sub Board (2019), the NIHR Health Technology Assessment (HTA) Clinical Evaluation and Trials Board (2007–9), the NIHR HTA Commissioning Board (2009–13), the NIHR Public Health Research Research Funding Board (2011–17), and the NIHR Programme Grants for Applied Research Expert Subpanel (2015–20). Naomi J Fulop and Steve Morris were supported in part by the NIHR Collaboration for Leadership in Applied Health Research Care (CLAHRC) North Thames at Barts Health NHS trust.

¹Department of Applied Health Research, University College London, London, UK

²Bristol Medical School, University of Bristol, Bristol, UK

³The Nuffield Trust, London, UK

⁴Department of Public Health and Primary Care, University of Cambridge, Cambridge, UK

^{*}Corresponding author n.fulop@ucl.ac.uk

Published October 2023 DOI: 10.3310/GQQV3512

Plain language summary

Rapid evaluation of the Special Measures for Quality and challenged provider regimes: a mixed-methods study

Health and Social Care Delivery Research 2023; Vol. 11: No. 19

DOI: 10.3310/GQQV3512

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

When health-care trusts in England have serious failings in the quality of care that they provide, they can be placed in 'Special Measures for Quality' and receive additional support from the NHS. There is also a list of 'challenged providers' at risk of entering Special Measures for Quality that receive support. In January 2019, of the 234 trusts in England, one-quarter had at some point been a challenged provider and/or entered Special Measures for Quality.

We studied how trusts responded to entering the Special Measures for Quality or challenged provider regimes. We wanted to understand if the support that the trusts receive can help the trust to improve the quality of care provided to patients. We did this by reviewing the relevant literature; speaking to a range of staff in eight trusts and nearby health organisations; analysing costs; and observing meetings in four of these trusts. We also compared national performance information between Special Measures for Quality/challenged provider trusts and non-Special Measures for Quality/challenged provider trusts.

We found that when a trust enters the Special Measures for Quality regime there is often an emotional impact on staff, who may experience low morale. Some staff thought that their trust received the right type of support, but others saw Special Measures for Quality as heavy-handed scrutiny or punishment. With hindsight, Special Measures for Quality was sometimes viewed more positively, as a pathway to make changes that were needed.

Looking at all trusts in England, we found that when trusts entered Special Measures for Quality or became challenged providers they started to get better at seeing emergency department patients within 4 hours and reduced avoidable deaths. We also found that some parts of the staff survey results improved.

We found that staff need time and space to make changes. Looking after staff and having a leadership and culture that supports continuous learning are important for making improvements. Regional health-care systems and local organisations have an important role to play in supporting trusts to make improvements.

Health and Social Care Delivery Research

ISSN 2755-0060 (Print)

ISSN 2755-0079 (Online)

Health and Social Care Delivery Research (HSDR) was launched in 2013 and is indexed by Europe PMC, DOAJ, INAHTA, Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA), NCBI Bookshelf and MEDLINE.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

This journal was previously published as *Health Services and Delivery Research* (Volumes 1–9); ISSN 2050-4349 (print), ISSN 2050-4357 (online)

The full HSDR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr.

Criteria for inclusion in the Health and Social Care Delivery Research journal

Reports are published in *Health and Social Care Delivery Research* (HSDR) if (1) they have resulted from work for the HSDR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HSDR programme

The HSDR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HSDR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-and-social-care-delivery-research.htm.

This report

The research reported here is the product of an HSDR Evidence Synthesis Centre, contracted to provide rapid evidence syntheses on issues of relevance to the health service, and to inform future HSDR calls for new research around identified gaps in evidence. Other reviews by the Evidence Synthesis Centres are also available in the HSDR journal.

The research reported in this issue of the journal was funded by the HSDR programme or one of its preceding programmes as project number NIHR129663. The contractual start date was in January 2019. The final report began editorial review in February 2020 and was accepted for publication in April 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HSDR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HSDR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the HSDR programme or the Department of Health and Social Care.

Copyright © 2023 Fulop *et al.* This work was produced by Fulop *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland, final files produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co)

RSET: The Rapid Service Evaluation Team

The Rapid Service Evaluation Team ('RSET'), comprising health service researchers, health economists and other colleagues from University College London and the Nuffield Trust, have come together to rapidly evaluate new ways of providing and organising care. We have been funded by the National Institute for Health and Care Research (NIHR) Health Service and Delivery Research (HS&DR) programme for five years, starting on April 1st 2018.

RSET are completing rapid evaluations with respect to:

- 1. The impact of services on how well patients do (e.g. their quality of life, how likely patients are to recover);
- 2. Whether services give people the right care at the right time;
- 3. Whether these services are good value for money;
- 4. how changes are put into practice, and what patients, carers, and staff think about how the changes happened and whether they think the changes made a difference;
- 5. What lessons there are for the rest of the NHS and care.

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board, Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk